Mauna Loa Solar Observatory Observer's Log _____ Mon Aug 24 17:11:14 GMT 2015 Year: 15 Doy: 236 Observer: berkey WEATHER COMMENT: berkey: Mon Aug 24 17:11:19 GMT 2015 Temp: 44.3f, Humidity: 92%, Pressure: 28.628in, Wind: 2mph from SE, Skies: Heavy overcast. end GONG COMMENT BY berkey: Mon Aug 24 17:12:20 GMT 2015 Gong station crashed yesterday and could not be recovered remotely. Tim thought it was GPS related; but the GPS appears t o be on and locked working. I will continue working with Tim to resolve this. end GENERAL COMMENT BY berkey: Mon Aug 24 19:08:18 GMT 2015 Talking to lots of folks about our new network upgrade. We should be switching over to UH from HawTel within 2 years. end GENERAL COMMENT BY berkey: Mon Aug 24 22:57:44 GMT 2015 Very thick fog now. end GONG COMMENT BY berkey: Mon Aug 24 22:58:24 GMT 2015 Tim was able to clear up the gong issues remotely after a reboot of the computer in the shelter. end
